Author's response to reviews

Title: HNF1B polymorphism influence the prognosis of endometrial cancer patients: a cohort study

Authors:

Vincenzo Dario Mandato (dariomandato@virgilio.it)
Enrico Farnetti (enrico.farnetti@asmn.re.it)
Federica Torricelli (federica.torricelli@asmn.re.it)
Martino Abrate (martino.abrate@asmn.re.it)
Bruno Casali (bruno.casali@asmn.re.it)
Gino Ciarlini (gino.ciarlini@asmn.re.it)
Debora Pirillo (debora.pirillo@asmn.re.it)
Mariacarolina Gelli (mariacarolina.gelli@asmn.re.it)
Mario Grassi (mario.grassi@unipv.it)
Davide Nicoli (davide.nicoli@asmn.re.it)
Giovanni B La Sala (giovannibattista.lasala@asmn.re.it)
Stefano Palomba (stefano.palomba@asmn.re.it)

Version: 5  Date: 13 March 2015

Author's response to reviews: see over
Dear Editor,

We would like to resubmit our revised manuscript entitled “HNF1B polymorphism influence the prognosis of endometrial cancer patients: a cohort study” for publication in *BMC Cancer.*

We revised our manuscript according to Editor suggestions. We hope that the new manuscript could be suitable for publication.

1) Requesting name of ethics committee: Please update your ethics statement to include the name of the ethics committee that approved your study

*Ethics*

*The Provincial Ethical Committee of Reggio Emilia* approved the study design and all patients provided written informed consent to use personal non-sensitive data at hospital admission.

2) Please provide funding statement in the acknowledgment section.

*Acknowledgements*

No fund or financial support has been received for this study

3) Please move the text currently in the acknowledgements to the competing interests section.

*Competing interests*

The authors declare that they have no competing interests. All authors deny any found, financial and personal relationships with other people or organizations/companies that could inappropriately influence their work.
We look forward to hearing from you as soon as possible.

Best Regards

Vincenzo Dario Mandato, MD, PhD

Department of Obstetrics and Gynecology,
Arcispedale S.Maria Nuova-IRCCS,
Viale Risorgimento 80, Reggio Emilia, Italy.

Phone:+393494640813 - fax: +390522296015;
e-mail: dariomandato@virgilio.it